\begin{table}%T4
\caption{List of the database reaction reference codes and their
correspondance with the bibliographical references. NOTE: Any digit-only code 
found in the database is the same code as that used for referencing 
reactions in Anicich (\cite{Anicich04}).} 
\label{codes}
%\centerline
 {
\begin{tabular}{llll@{\qquad}ll}
\hline \hline
\noalign{\smallskip}
Code & References & Code & References & Code & Reference\\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
AA83 & Alge et~al. (\cite{Alge83}) & AB93 & Abouelaziz et~al. (\cite{Abouelaziz93}) & AM06 & Andreazza et~al. (\cite{Andreazza06}) \\
AS95 & Andreazza et~al. (\cite{Andreazza95}) & AS84 & Adams et~al. (\cite{Adams84}) & AS97 & Andreazza \& Singh (\cite{Andreazza97}) \\
 AS04 & Asvany et~al. (\cite{Asvany04}) & BA84 & Barlow (\cite{Barlow84}) & BH95 & Bettens \& Herbst (\cite{Bettens95})    \\
 BS97 & Brownsword et~al. (\cite{Brownsword97}) & BV06 & Barinovs \& van Hemert (\cite{Barinovs06}) & CA01 & Canosa et~al. (\cite{Canosa01}) \\
 CG06 & Carty et~al. (\cite{Carty06}) & CH94 & Clary et~al. (\cite{Clary94}) & DA97 & Decker \& Adams (\cite{Decker97}) \\
 DD90 & Dalgarno et~al. (\cite{Dalgarno90}) & EH04 & Ehlerding et~al. (\cite{Ehlerding04})  & FA80 & Field et~al. (\cite{Field80}) \\
 FM01 & Fikri et~al. (\cite{Fikri01}) & FO02 & Fockenberg \& Preses (\cite{Fockenberg02}) & GA04 & Geppert et~al. (\cite{Geppert04b}) \\
 GA06 & Geppert et~al. (\cite{Geppert06})       & GB04 & Geppert et~al. (\cite{Geppert04c}) & GC04 & Geppert et~al. (\cite{Geppert04d}) \\
 GE05 & Montaigne et~al. (\cite{Montaigne05}) & GH04 & Geppert et~al. (\cite{Geppert04a}) & GL87 & Gredel et~al. (\cite{Gredel87}) \\
 GL89 & Gredel et~al. (\cite{Gredel89})        & GT04 & Geppert et~al. (\cite{Geppert04g}) & GW04 & Geppert et~al. (\cite{Geppert04f}) \\
 HD89 & Herbst et~al. (\cite{Herbst89a}) & HE85 & Herbst (\cite{Herbst85}) & HE87 & Herbst (\cite{Herbst87}) \\
 HE06 & Heber et~al. (\cite{Heber06}) & HH93 & Haider \& Husain (\cite{Haider93}) & HL86 & Herbst \& Leung (\cite{Herbst86a}) \\
 HL89 & Herbst \& Leung (\cite{Herbst89}) & HL90 & Herbst \& Leung (\cite{Herbst90}) & HM89 & Herbst et~al. (\cite{Herbst89b})\\
 HT00 & Herbst et~al. (\cite{Herbst00}) & IS89 & Smith (\cite{Smith89})  & JE00 & Jensen et~al. (\cite{Jensen00}) \\
  KA02 & Kalhori et~al. (\cite{Kalhori02}) & KD93 & Kimura \& Dalgarno (\cite{Kimura93})  & KS96 & Kaiser et~al. (\cite{Kaiser96}) \\
 LG88 & Leen \& Graff (\cite{Leen88}) & LL98 & Larson et~al. (\cite{Larson98}) & LU02 & Luca et~al. (\cite{Luca02}) \\
 MB73 & McGregor \& Berry 1973 & MB87 & Millar et~al. (\cite{Millar87}) & MC04 & McCall et~al. (\cite{McCall04}) \\
 MH90 & Millar \& Herbst (\cite{Millar90}) & MH91 & Millar et~al. (\cite{Millar91}) & MI84 & Mitchell (\cite{Mitchell84}) \\
 MI90 & Mitchell (\cite{Mitchell90}) & MI91 & Millar (\cite{Millar91}) & MM93 & Maluendes et~al. (\cite{Maluendes93}) \\
 MS99 & McEwan et~al. (\cite{McEwan99}) &NA99 & Nahar (\cite{Nahar99}) & NIST & NIST Chemical Kinetics Database \\
 NM90 & Nesbitt et~al. (\cite{Nesbitt90}) & NM05 & Novotny et~al. (\cite{Novotny05}) & NP97 & Nahar \& Pradhan (\cite{Nahar97})\\
 NW05 & Neufeld et~al. (\cite{Neufeld05})  & PD89 & Petuchowski et~al. (\cite{Petuchowski89}) & PE98 & Peterson et~al. (\cite{Peterson98}) \\
 PH80 & Prasad \& Huntress (\cite{Prasad80}) & PM03 & Petrie et~al. (\cite{Petrie03}) & RA92 & Rawlings (\cite{Rawlings92}) \\
 RI99 & Rim \& Hershberger (\cite{Rim99})  & RJ91 & Roberge et~al. (\cite{Roberge91}) & RO00 & Rosen et~al. (\cite{Rosen00}) \\
 SA00 & Singh \& Andreazza (\cite{Singh00}) & SD98 & Stancil \& Dalgarno (\cite{Stancil98}) & SE98 & Semaniak et~al. (\cite{Semaniak98}) \\
 SE01 & Semaniak et~al. (\cite{Semaniak01}) & SH04 & Smith et~al. (\cite{Smith04}) & SI99 & Singh et~al. (\cite{Singh99}) \\
 SI06 & Sims (\cite{Sims06}) & SL98 & Stancil et~al. (\cite{Stancil98}) & SS94 & Smith et~al. (\cite{Smith94}) \\
 SS99 & Stancil et~al. (\cite{Stancil99}) & ST00 & Stancil et~al. (\cite{Stancil00}) & SY92 & Suzuki et~al. (\cite{Suzuki92}) \\
 TE96 & Talbi et~al. (\cite{Talbi96}) & VA99 & Vikor et~al. (\cite{Vikor99}) & VD87 & van Dishoeck (\cite{Dishoeck87}) \\
 VD88 & van Dishoeck (\cite{Dishoeck88}) & VD06 & van Dishoeck et~al. (\cite{Dishoeck06}) & VP83 & Viggiano \& Paulson (\cite{Viggiano83}) \\
 WB88 & Wlodek et~al. (\cite{Wlodek88}) & WG04 & Geppert et~al. (\cite{Geppert04e}) & ZS98 & Zygelman et~al. (\cite{Zygelman98})\\
\noalign{\smallskip}
\hline
\end{tabular}}
\smallskip
The NIST Chemical Database is available on-line at {\tt http://kinetics.nist.gov/index.php}
\end{table}